Author's response to reviews

Title: Two novel missense mutations in the myostatin gene identified in Japanese patients with Duchenne muscular dystrophy

Authors:

Atsushi Nishiyama (034d595m@y03.kobe-u.ac.jp)
Yasuhiro Takeshima (takesima@med.kobe-u.ac.jp)
Kayoko Saiki (ksaiki@med.kobe-u.ac.jp)
Akiko Narukage (ped@med.kobe-u.ac.jp)
Yoshinobu Oyazato (oyazato@med.kobe-u.ac.jp)
Mariko Yagi (myagi@med.kobe-u.ac.jp)
Masafumi Matsuo (matsuo@kobe-u.ac.jp)

Version: 4 Date: 17 March 2007

Author's response to reviews: see over
Dear Dr. Norton,

It is my great pleasure to hear that our manuscript (MS: 8093524641197856) is accepted for publication in BMC Medical Genetics after revision. I appreciate all suggestions and comments from two reviewers. Therefore I corrected our manuscript based on the reviewer’s comments. In addition several points that were not pointed out by reviewers were also corrected.

I am sending our re-revised manuscript entitled “Two novel missense mutations in the myostatin gene identified in Japanese patients with Duchenne muscular dystrophy” for the publication in BMC Medical Genetics. On behalf of all co-authors I declare that this is an original manuscript and not under the review for any other journal.

The followings are answers to comments from each reviewer;

Reviewer 1
One sentence in page 12 was mistyped and corrected. To make the description clear sentences in page 11 and 12 were revised.

Reviewer 2
The suggested points were corrected.

I hope this revised manuscript is accepted for publication.

Sincerely yours

Masafumi Matsuo MD PhD
Professor and Chairman
Department of Pediatrics
Kobe University Graduate School of Medicine
7-5-1, Kusunoki, Chuo, Kobe 650-0017 Japan
Tel: +81-78-382-6080 Fax: +81-78-382-6098 E-mail: matsuo@kobe-u.ac.jp